Author’s response to reviews

Title: Hepatitis C Virus Screening and Treatment in Irish Prisons from a Nurse Managers' Perspectives - A Qualitative Exploration

Authors:

Des Crowley (doctordes@hotmail.com)
Marie Claire Van Hout (M.C.VanHout@ljmu.ac.uk)
Carol Murphy (murphc95@tcd.ie)
Enda Kelly (ETKelly@irishprisons.ie)
John Lambert (jlambert@mater.ie)
Walter Cullen (walter.cullen@ucd.ie)

Version: 2 Date: 24 Apr 2019

Author’s response to reviews:

24th April 2019

NURS-D-18-00133R1
Hepatitis C Virus Screening and Treatment in Irish Prisons from a Nurse Managers' Perspectives - A Qualitative Exploration

Dear Editor,

Please find attached a revised manuscript, with the following revisions related to your submission response:

1. See below revisions regarding overlap between this manuscript and other sources:

Original: The healthcare of prisoners is an integral aspect of incarceration

Revised: Prison healthcare is an essential aspect of incarceration
IPS nurses have a broad range of professional experience and qualifications.

Their clinical work is diverse and includes multiple areas of responsibility.

Peer interventions need to be incorporated into prison life and require a collaborative approach between prison health care providers and prison services to be delivered effectively.

2. The abstract has been revised to include the required headings.
3. Clarification has been provided regarding written consent: “All study participants provided written informed consent.”
4. A list of abbreviations is now included after the conclusion section.

Yours Sincerely,
Des